Author's response to reviews

Title: miR-320b suppresses cell proliferation by targeting c-Myc in human colorectal cancer cells

Authors:

Hantao Wang (hantaowang@126.com)
Fuao Cao (fuaocao2003@163.com)
Xu Li (xuli_ch@163.com)
Hua Miao (13567330917@139.com)
Jifu E (jefershon@sohu.com)
Junjie Xing (xjj2118@163.com)
Chuan-gang Fu (fugang416@126.com)

Version:6 Date:14 June 2015

Author's response to reviews: see over
Dear Ryan Relox:

We appreciate your time reviewing our manuscript MS: 2079497320154697 - miR-320b suppresses cell proliferation by targeting c-Myc in human colorectal cancer cells.

We have studied editor’s comments carefully and have made revision which marked in yellow in the paper. We have provided the point-by-point responses below, and we think we have addressed all the concerns.

We would like to express our great appreciation to you for comments on our paper. Looking forward to hearing from you.

Thank you very much for your consideration.

Yours sincerely,

Chuan-gang Fu, M.D.
Professor and Director, Department of Colorectal Surgery, Changhai Hospital, Shanghai 200433, China
Email: fugang416@126.com

Point-to-point responses

Responses to request # 1: Keywords to be included after the Abstract.

Response: Many thanks for the positive comments. According to the comments, we have added the keywords. We added the related description in the manuscript (Page 3, Line 5).

Responses to request # 2: Clarify in the Methods if consent was written or oral.

Response: Many thanks for the positive comments. According to the comments, we have corrected the oral consents. We added the related description in the manuscript (Page 8, Line 3, 14, 18-19; Page 9, Line 2).